Author's response to reviews

Title: "A powerful intervention: General Practitioners' use of sickness certification in depression"

Authors:

Sara M Macdonald (sara.macdonald@glasgow.ac.uk)
Maragaret Maxwell (margaret.maxwell@stir.ac.uk)
Philip Wilson (phil.wilson@glasgow.ac.uk)
Michael Smith (michael.smith3@nhs.net)
William Whittaker (william.whittaker@manchester.ac.uk)
Matt Sutton (matt.sutton@manchester.ac.uk)
Jillian Morrison (jill.morrison@glasgow.ac.uk)

Version: 4 Date: 30 July 2012

Author's response to reviews:

Dear Editor

Many thanks for allowing us to resubmit our paper and for the helpful reviewer comments, in particular that they were happy with the changes made. We have provided a response to each of the reviewer’s comments, below.

Reviewer 1

The authors have made good improvements to the references, but there are still a few minor mistakes in them. A few missing spaces, a bold faced publication year etc.

We have looked once again at the references and have corrected the errors and hope that the references now fully adhere to the house style.

Yours sincerely

Sara Macdonald